Chohan /21--Nov-05
Week 47 starting  21 Nov 05
SM18 Operation Meeting (Tuesdays)

Bench status on Monday 21 Nov 05:  
 11/21/2005 2:30:21 PM 
 
	7
	A1
	MBAR2226
	ICS 4 Final connection
	291.6

	1
	A2
	MBAL3378
	PT 12 Warm Up
	6.03

	8
	B1
	SSS149
	ICS 3 Traveller after warm test
	291.6

	5
	B2
	SSS147
	PT 13.2 Resist. Meas.
	298.04

	2
	C1
	MBAR2249
	PT 12 Warm Up
	281.79

	3
	C2
	MBAR2246
	Prep 5 Cool Down
	88.17

	9
	D1
	SSS208
	Prep 1 Lyre Test
	290.71

	10
	D2
	SSS289
	PT 13.2 Resist. Meas.
	293.57

	4
	E1
	MBAL1088
	Prep 5 Cool Down
	101.94

	12
	E2
	MBBR3388
	Prep 0 Scan Bench
	291.96

	6
	F1
	MBBL1192
	Prep 5 Cool Down
	289.1

	11
	F2
	MBBR3339
	ICS 2 Connect Magnet (ICS)
	290.17


 
   
Coordinator this week: Nikolay; MTM support: 16-3602

 18 complete Magnet Tests cycles  12 Dipoles  + 6 SSS + 0 SSSS

 Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight)
  [ 2140 ROBTC ; sss277 3rd time; 2179 2nd time ]
Situation From Cryogenics:  J-P Lamboy or ACR Rep.

Special Points:
· Lots of quenches  on SSS277 ?

Results from 3139 ?

What was wrong with 2179 ?

Next Special SSS Q6 on D2 (J Halik  , Marco)  SSW @warm to start
AOB 
	Magnets under Test from 13-11-2005 to 19-11-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for

	1
	TBA1
	MBAL1093 
	[image: image1.png]



	


	10-11-2005 
	33 h.
	27 h.
	18 h.
	13 h.
	8 h.
	14-11-2005 
	99 h.
	Stripping

	2
	TBA2
	MBAL3398 
	[image: image3.png]



	


	12-11-2005 
	29 h.
	21 h.
	18 h.
	15 h.
	10 h.
	16-11-2005 
	93 h.
	Stripping

	3
	TBB1
	SSS277 
	[image: image5.png]



	


	08-11-2005 
	30 h.
	15 h.
	93 h.
	10 h.
	29 h.
	15-11-2005 
	177 h.
	Stand By

	4
	TBB2
	SSS157 
	[image: image7.png]



	


	09-11-2005 
	32 h.
	24 h.
	18 h.
	7 h.
	36 h.
	14-11-2005 
	117 h.
	Stripping

	5
	TBB2
	SSS225 
	[image: image9.png]



	


	14-11-2005 
	31 h.
	11 h.
	15 h.
	8 h.
	12 h.
	17-11-2005 
	77 h.
	Stripping

	6
	TBC1
	MBAL1181 
	[image: image11.png]



	


	09-11-2005 
	28 h.
	31 h.
	14 h.
	16 h.
	15 h.
	14-11-2005 
	104 h.
	Stand By

	7
	TBC1
	MBAR3328 
	[image: image13.png]



	


	14-11-2005 
	24 h.
	19 h.
	20 h.
	14 h.
	9 h.
	17-11-2005 
	86 h.
	Stripping

	8
	TBC2
	MBAL3414 
	[image: image15.png]



	


	11-11-2005 
	26 h.
	21 h.
	14 h.
	19 h.
	17 h.
	15-11-2005 
	97 h.
	Stripping

	9
	TBC2
	MBBR3364 
	[image: image17.png]



	


	16-11-2005 
	25 h.
	21 h.
	13 h.
	15 h.
	9 h.
	19-11-2005 
	83 h.
	Stripping

	10
	TBD1
	SSS123 
	[image: image19.png]



	


	11-11-2005 
	36 h.
	13 h.
	14 h.
	8 h.
	15 h.
	15-11-2005 
	86 h.
	Stripping

	11
	TBD2
	SSS179 
	[image: image21.png]



	


	10-11-2005 
	27 h.
	24 h.
	22 h.
	6 h.
	15 h.
	14-11-2005 
	94 h.
	Stripping

	12
	TBD2
	SSS193 
	[image: image23.png]



	


	14-11-2005 
	29 h.
	18 h.
	16 h.
	10 h.
	10 h.
	17-11-2005 
	83 h.
	Stripping

	13
	TBE1
	MBAR2242 
	[image: image25.png]



	


	14-11-2005 
	47 h.
	21 h.
	21 h.
	14 h.
	10 h.
	19-11-2005 
	113 h.
	Stripping

	14
	TBE1
	MBBL2179 
	[image: image27.png]



	


	07-11-2005 
	31 h.
	21 h.
	54 h.
	16 h.
	45 h.
	14-11-2005 
	167 h.
	Stand By

	15
	TBE2
	MBBL2140 
	[image: image29.png]



	


	07-11-2005 
	65 h.
	25 h.
	59 h.
	-
	-
	14-11-2005 
	149 h.
	Not Found or Thermal cycle

	16
	TBE2
	MBBL2140 
	[image: image31.png]



	


	14-11-2005 
	0
	23h
	10h-
	22h-
	10h-
	17-11-2005 
	65 h.
	Stripping

	17
	TBF1
	MBAR2232 
	[image: image33.png]



	


	09-11-2005 
	28 h.
	23 h.
	25 h.
	16 h.
	25 h.
	14-11-2005 
	117 h.
	Stripping

	18
	TBF2
	MBBR1281 
	[image: image35.png]



	


	12-11-2005 
	29 h.
	20 h.
	18 h.
	14 h.
	7 h.
	16-11-2005 
	88 h.
	Stripping


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11

	1
	MBAL1093
	11582.3 A.
(D1 - L)
	12542.4 A.
(D2_U)
	12600.2 A.
(D1_L)
	[image: image37.png]



	-
	-
	-
	-
	-
	-
	-

	2
	MBAL3398
	11789.4 A.
(D1_U)
	12850 A.
[image: image38.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS277
	10313 A.
(QI_34)
	10952 A.
(QI-34)
	11188 A.
(QI-34)
	11027 A.
(QI-34)
	11286 A.
(QI-34)
	11427 A.
(QI-34)
	11089.8 A.
(QI34IC4)
	11522.7 A.
(QE_12_)
	11610 A.
(QI-34)
	11353 A.
(QI-34)
	11312 A.
(QI-34)

	3
	SSS277
	11307.5 A.
(QI_34)
	11549.4 A.
(QI_34)
	11601 A.
(QI_12)
	11570 A.
(QI_34)
	11157.7 A.
(QI-34-C4)
	
	
	
	
	
	

	4
	SSS157
	11801.7 A.
(QI-34)
	12004.6 A.
(QE-12)
	12370.2 A.
(QI_12)
	[image: image39.png]



	-
	-
	-
	-
	-
	-
	-

	5
	SSS225
	12583.9 A.
(QI_34-12)
	12850 A.
[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBAL1181
	12226.4 A.
(D2_L)
	12433 A.
(D2 - L)
	12777.1 A.
(D2_L)
	[image: image41.png]



	-
	-
	-
	-
	-
	-
	-

	7
	MBAR3328
	11571.4 A.
(D1-L)
	11871.7 A.
(D1-U)
	12227.3 A.
(D2_L)
	12850 A.
[image: image42.jpg]



	-
	-
	-
	-
	-
	-
	-

	8
	MBAL3414
	12698 A.
(D2-L)
	12850 A.
[image: image43.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBR3364
	12757 A.
(D1_L)
	12849 A.
[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS123
	12070.8 A.
(QI - 12)
	12261 A.
(QE-12)
	12400 A.
(QI-12)
	[image: image45.png]



	-
	-
	-
	-
	-
	-
	-

	11
	SSS179
	12112.6 A.
(QE-34)
	12126.6 A.
(QI_12)
	12449.8 A.
(QI-34)
	[image: image46.png]



	-
	-
	-
	-
	-
	-
	-

	12
	SSS193
	11852.9 A.
(QI-12)
	12443.3 A.
(QI-34)
	12670.3 A.
(QI-12)
	[image: image47.png]



	-
	-
	-
	-
	-
	-
	-

	13
	MBAR2242
	11992.4 A.
(D1-U)
	12849.4 A.
(D1 - U)
	12856 A.
[image: image48.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBBL2179
	12564.4 A.
(D1_U)
	11790.2 A.
(D1-U)
	12338.1 A.
(D1-U)
	11846.9 A.
(D1-U)
	12473 A.
(D1-U)
	12649 A.
(D1-U)
	12700.8 A.
(D1_L)
	[image: image49.png]



	-
	-
	-

	15
	MBBL2140
	10858.8 A.
(D1-U)
	11943.3 A.
(D2-L)
	12705 A.
(D1_U)
	11894.7 A.
(D1_U)
	12771.3 A.
(D1_U)
	12218.5 A.
(D1-U)
	12255.1 A.
(D1-U)
	12315 A.
(D1_U)
	12710.7 A.
(D1_U)
	[image: image50.png]



	-

	16
	MBBL2140
	12653.9 A.
(D1-U)
	12850 A.
[image: image51.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	17
	MBAR2232
	10905.7 A.
(D1_U)
	11920.7 A.
(D2_L)
	12850 A.
[image: image52.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	18
	MBBR1281
	11369 A.
(D1_U)
	12276.5 A.
(D2_L)
	12297.5 A.
(D1-U)
	[image: image53.png]



	-
	-
	-
	-
	-
	-
	-
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2140 ROBTC ;  2179 2nd time round (Vjump) & lots of tests but magnet accepted anyway; SSS277  3rd time & also why so many quenches?
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Wednesday_16_M
	TBC1


	Power cut. Storage down. First training quench for MB3328 was analysed in local sun.
	
	
	

	Wednesday_16_A
	TBC1


	Training quench 3 on MB3328: bump observed in the current discharge indicating that the diode is conducting.
	AB/PO?
	
	Guy informed. Delay in the FPA? Power converter is checked.

	Wednesday_16_A
	TBB1


	Errors in voltage taps signals displayed during IAP at warm on SSS161.
	MTM
	
	

	Wednesday_16_N
	TBD1


	Leak on SSS165.
	ICS?
	
	Magnet re-opened.

	Thursday_17_M
	TBF1


	TT821 value not correct seen in MB3139.
	MTM
	
	Guy informed. He found that one connection V+ is opened. NC opened.

	Thursday_17_A
	TBA1


	MM2 on MB2201: rms of the rotation speed is above 2% on aperture 2.
	MTM
	
	Guy informed Stephane. He advised to continue with load-line and asked to verify if during the measurements the situation degrades.

	Thursday_17_N
	TBA2


	MB3378: Leak found in a flexible during magnet connection. 
	ACR
	
	Wait up to the morning to repair.

Solved Friday mid day?

	Thursday_17_N
	TBA1


	HV test final @4.5 K on MB2201 could not be carried out immediately because of bad temperature regulation.
	ACR
	
	

	Friday_18_M
	TBB1


	SSS161:QH resistance measurement at warm: offset value slightly above the threshold with one multi-meter (0.173 ohms) of the SSS measurement rack.
	MTM
	
	Multi-meter changed  (for the dipole) and offset in specification.

	Friday_18_M
	TBD1


	Powering of corrector IJ for the SSS165: test failed because a PC fault appeared when the current is ramped.
	AB/PO?
	
	Guy checked. The test passed after the changing of the power supply.  Control of the power supply asked to AB/PO people.

	Saturday_19_N
	TBC1


	MB2249: Alerts during PT 5.2 (HF heater provoked quench): QH1,3 tminV<500 and QH3 Vmin>15.0
	MTM
	AQA
	

	Sunday_20_M
	all


	Cryo OK lost for all magnets at 1.9K. Liquefactor down.
	ACR
	
	OK after 5 hours.

	Sunday_20_M
	TBC1


	MB2249: Alerts on Minv and TminV for QH3
	MTM
	AQA
	OK by TC and Guy after checks.

	Sunday_20_A
	TBC1


	MB2249:Noise in Triggers & current in quench no.1 PT6.1.
	MTM
	AQA
	OK by TC as there is no alert in AQA. If the noisy signals persist in the second quench he advised to stop the training and perform HV test.

	Sunday_20_A
	TBD1


	SSS165: Wrong inductance value for the circuit EF (MO) in the TEMA setting.
	MTM
	
	Inductance value of 3mH set manually and test performed. The setting has to be corrected.


